Occupations over the business cycle

Péter Hudomiet

April 29, 2013

University of Michigan

JEL codes: J24, J64

PRELIMINARY AND INCOMPLETE. PLEASE DO NOT CITE.¹

Abstract

The business cycle does not have a homogenous effect on workers in different occupations. This paper documents and analyzes unemployment patterns between 1983 and 2012 in broad and detailed occupation categories using the CPS. I find that the skill level of occupations has a monotone negative effect on the level and volatility of unemployment. This result is robust to different measures of general skills and it occurred in all recessions in last 30 years, including the 2007 one. The result also holds within and across major occupation categories. In order to understand the causes of it, I decompose unemployment fluctuations into changes in the job-loss and the job-finding probabilities in each occupation. I find that even though separations are less important for aggregate fluctuations in unemployment, they are more important for understanding the differences between occupational unemployment, as the job-finding probability follows a highly similar pattern across occupations. These results are in line with a model in which employers are more likely to hoard skilled workers due to higher recruitment/training and other adjustment costs in skilled jobs. I consider alternative models and argue that none can explain why job-finding probabilities are so similar while unemployment rates are so different across occupations. Particularly, it seems unlikely that productivity differences are the main driving fource behind occupational unemployment patterns. Future research will compare the evolution of occupational job-loss and job-finding probabilities in the US to those in Germany, and I plan to propose a new labor protection system which maximizes the welfare of citizens by concentrating protection in risky occupations and uncertain times and therefore minimizing the incidence of unemployment and decreasing the burden on the unemployment insurance systems.

¹Any comments would be highly appreciated at hudomiet@umich.edu.

1 Introduction

Numerous studies have shown that the effect of recessions on different demographic groups is heterogeneous (Clark and Summers, 1981; Kydland, 1984; Keane and Prasad, 1993; Hoynes, 1999; Jaimovich and Siu, 2009 and Hoynes et al., 2012). Unskilled workers (the young, the less educated, some minority groups, the recently employed) are found to exhibit greater fluctuations in unemployment and smaller fluctuations in their income. These findings are usually interpreted as mainly due to some sort of "skill" differences among these groups. However, little is known about what skills are actually rewarded in recessions and why. This paper documents in detail the relationship of occupational skills and unemployment transition probabilities across the business cycle and examines empirically the predictions arising from a comprehensive set of candidate explanations.

First, I examine which jobs are risky and which jobs are safe in recessions. I create a consistent and detailed labor force categorization into 191 occupations and track how within-occupation rates of unemployment evolved in the last 30 years. I find very big differences between the level and the volatility of unemployment rates across occupations. For example, while the unemployment rate among highly skilled professional workers went up from 1.7 percent in 2006-2007 to 3.5 percent in 2009-2012, it went up from 8 percent to 15 percent among low skilled operators and laborers. By looking at detailed occupation categories, I find that the skill level of occupations has a strong negative effect on unemployment prospects in and after recessions. This finding is robust to the definition of skills in occupations (for example, average years of education or wage in different census years; non-routine cognitive skills using the O*NET data), and to different recessions, including the 2007 one. When occupational skills are defined as average years of completed education, the results are stronger and monotone, with the lowest skilled occupations being the most risky. For example, while the unemployment rate in the lowest educated decile went up from 9.5 percent in 2006-2007 to 15.4 percent in 2009-2012, it went up from 1.3 percent to 2.8 percent in the highest educated decile. When occupational skills are defined as average log wages, or various direct skill measures (cognitive, psychomotor or physical) the results are somewhat weaker.

Secondly, I investigate the source of the differential unemployment patterns in occupations. I suggest simple tests of six potential explanations. I compute job-loss and job-finding probabilities across occupations over time. This approach is motivated by two observations. Fluctuations in the aggregate unemployment rate is strongly influenced by changes in the job-finding probability while the job-loss probability has a smaller and less persistent effect (Shimer, 2012; Davis et al., 2006; Elsby et al., 2009; Fujita and Ramey, 2009). In contrast, Elsby et al. (2010) find that fluctuations in the job-finding probability are remarkably similar across different demographic groups while the job-loss probability varies greatly. I repeat this exercise with my detailed occupation classification and find similar results: even though the job-finding probability is highly pro-cyclical, it follows broadly the same pattern in all occupational groups. This is not the case for the job-loss probability: even though it is only weakly counter-cyclical, there is a big variation across occupations, with the unskilled ones being more volatile and higher in levels. These findings question the conventional premise that some skills are particularly productive in recessions. Instead, they are consistent with a model in which employers are more likely to hoard skilled workers. Alternative models predict that the job-finding probability is more stable in skilled occupations. However, this is not observed in the data. Out of the six hypotheses I consider, I find labor hoarding to be the most important.

The literature provides two main rationales for hoarding labor: the existence of adjustment costs and firm specific skills. Thus, skilled workers might be hoarded more either because hiring, training and other adjustment costs are higher or because skilled workers possess more firm specific skills.² Unfortunately, empirically we know very little about the importance of these factors; how they vary by occupations and perhaps over time. As Oyer and Schaefer (2011) noted in their chapter in the last Handbook of Labor Economics "We are certainly not the first to make the point that the demand side of the labor market—and hiring, in particular—is understudied." Nevertheless, the available evidence confirms that hiring costs, particularly training costs, are substantial and that they are proportionally higher for skilled workers. (See Manning (2011) section 2.1.2 for a review of this scarce literature.)

Understanding recruitment costs, thus, is very important to understand labor market dynamics in different occupations. The structure of these costs, however, is quite complex. In an interesting paper Blatter et al. (2012) estimated that roughly 70 percent of total hiring costs are related to what they call *adaptation* costs, mostly reduced productivity of new hires, which was missing from previous estimates. The literature on employer training (Lillard and Tan, 1986; Brown, 1990; Lerman et al., 2004; Leuven, 2005 provide thorough reviews) also find that training is widespread and it correlates positively with

 $^{^{2}}$ There is also a somewhat related literature on implicit contracts (e.g. Baily, 1974, Azariadis, 1975, 1976). The theory assumes that firms are risk-neutral, workers are risk-averse and they can write long-term contracts about wage and employment fluctuations contingent on future product demand conditions. The main result of this theory is that optimal contracts specify rigid wages as a form of insurance provided by the firm to the worker. Employment, however, is more likely to be stochastic, but under some conditions full employment is also part of the insurance.

workers' skills. Moreover, they also emphasize the difficulty in measuring informal training which is perhaps more important than formal training.

Understanding the source of variation in occupational unemployment is important for several reasons. First, many studies have shown that unemployment has a major negative effect on the welfare of affected families (Jacobson et al., 1993; von Wachter et al., 2009; Hijzen et al., 2010). It is important to know how this risk is concentrated in certain job types. A related second point is that labor economists should care about occupational unemployment, because workers seem to care, too. If one writes "recession" in Google then one of the first pop-up search offers is "recession proof jobs". By choosing it one gets almost 3 million hits with numerous popular websites, like Wikipedia or Time Business, offering a list of best recession proof jobs. It seems to be a common knowledge that there are more and less risky jobs and one should aim for choosing the latter ones. Thus, it is surprising how little labor economists know about the extent and determinants of this risk. Third, my paper is important for its policy implications. I lay out a simple rigid wage model with hiring and firing costs and show that in occupations where adjustment costs are low, the market outcome features inefficiently high lay-off rates in recessions. Therefore, a policy that increases firing costs, through for example taxing lay-offs, in these occupations during weak economic conditions can be welfare improving, because it minimizes the incidence of inefficient unemployment. Another policy implication is that supporting job-creation in recessions is ineffective and wasteful, because firms already hold excess labor in recessions. The last reason why my results are useful is that by analyzing the heterogeneity in job-loss, job-finding, and unemployment in occupations we learn about the hiring strategies of firms which has been an understudied area in labor economics.

My paper is related to several different literatures. The skill content of occupations became an important factor in explaining recent trends in wage inequality and job polarization (Autor et al., 2003; Autor et al., 2006; Goos and Manning, 2007; Firpo et al., 2011; Acemoglu and Autor, 2011); job mobility and task specific human capital (Ingram and Neumann, 2006; Poletaev and Robinson, 2008; Abraham and Spletzer, 2009; Kambourov and Manovskii, 2009; Geel and Backes-Gellner, 2009; Gathmann and Schönberg, 2010; Robinson, 2011); and learning and career decisions (James, 2011; Yamaguchi, 2012). The business cycle properties of occupations, however, are rarely analyzed. Moscarini and Vella (2008) find that occupation choice appears to be more "random" in recessions. Devereux (2000, 2004) finds that people get lower quality jobs in recessions and this explains half of the wage procyclicality of new hires. Campbell III (1997) found that wages are more rigid in skilled 1 digit occupations. These results are in line with and complementary to the ones of this paper. Jaimovich and Siu (2012) show that the permanent disappearance of some middle skilled occupations took place mostly in downturns. Acemoglu and Autor (2011) discuss that middle skilled occupations continued to disappear in the 2007 recession. Even though low skilled occupations have not been disappearing in the last 30 years, I will show that they were more strongly affected by the business cycle in the short run compared to middle skilled occupations.

My paper is also related to the macro-labor literature of the business cycle with heterogeneous labor. A couple of papers derived search and matching models with skilled and unskilled labor and jobs (Acemoglu, 1999; Albrecht and Vroman, 2002; Dolado et al., 2009; Khalifa, 2009; Chassamboulli, 2011). One implication of these models is that skilled workers might crowd out the unskilled in recessions and move back into skilled jobs in booms. They also find some empirical evidence for this mechanism. I will show, however, that it explains little of the overall differential between the unemployment patterns of occupations. Some papers find evidence for rising hiring standards and quality adjustment of labor over the cycle (Reder, 1955; Teulings, 1993; van Ours and Ridder, 1995; Solon et al., 1997; Evans, 1999; Devereux, 2004; Büttner et al., 2010), although it does not appear to be very strong.

My paper is also related to the literature on labor hoarding. Since the seminal work of Oi (1962), many papers found strong evidence for labor hoarding (for example Fay and Medoff, 1985; Fair, 1985; Burnside and Eichenbaum, 1996; Basu and Kimball, 1997; Marchetti and Nucci, 2001 and Liu and Spector, 2005) and analyzed the causes of it (for example Miller, 1971; Topel, 1982; Hamermesh, 1995; Galeotti et al., 2005; Wen, 2005 and Platt and Platt, 2011). These papers rarely analyze the occupational differences in labor hoarding. The original Oi (1962) paper is one exception, who showed that recruitment costs are lower and labor turnover is higher among "Common laborers".

The paper is organized as follows. Section 2 shows patterns in unemployment, job-loss and job-finding probabilities in detailed occupations and shows the relationship between these patterns and various measures of occupational skills. Section 3 discusses and tests a set of potential economic explanations, and it concludes that large differences in the extent of labor hoarding is the only explanation that is consistent with all the available the data. Section 4 describes my simple labor hoarding model and discusses its policy implications and Section 5 concludes.

2 Descriptive analysis of the business cycle properties of occupations

In this section I show properties of occupational unemployment over the business cycle. In the first half, I analyze how volatile unemployment is within occupations, and how it varies with skills. In the second half of the section, I compute job-loss and job-finding probabilities within occupations and show how they vary by skills and over time.

There is no unique way of defining skill levels of occupations in the literature. There are two main decisions to make: 1. how detailed occupation classification and; 2. what proxy of skills we should use. My decision for the first one was to use 3 different levels of aggregation. The most aggregate one follows the definition of Acemoglu and Autor (2011) and uses ten occupation categories.^{3,4} Some contain mostly high-paid high skilled white-collar occupations; some are middle skilled white-collar ones; and some are mixes of middle and low skilled blue-collar or service jobs. I will show, that there is an enormous variation in the unemployment prospects within these occupation categories and it is worth defining occupations at a finer level. The most detailed occupation categorization I use is the 383 three digit 1990 census codes and the crosswalk created by IPUMS.⁵ This classification turns out to be too detailed, as some cells have too few people in them, and some occupations even disappear over time. As a remedy I created a new classification and cross walk for 191 occupations.⁶ Results based on the 383 and the 191 occupations turn out to be almost identical.

Second, I had to choose a proxy for occupational skills. There are three approaches in the literature. One is to directly use the major occupational groups; the second is to use average wage or average years of education as proxies; and the third is to use an outside data-source (like the Dictionary of Occupational Titles or the O*NET) to merge task content of occupations. The O*NET⁷ has a very detailed evaluation of more than 1000 occupations in dimensions like abilities and skills of workers, work styles, worker requirements, tasks, etc. My preferred definition is the average years of education

³These are 1. management and related; 2. professionals; 3. technicians; 4. sales; 5. office and administrative support; 6. production, craft and repair; 7. operators, fabricators and laborers; 8. protective services; 9. food preparation and cleaning and 10. personal care and personal services.

⁴Acemoglu and Autor (2011) convincingly show that there has been a trend of disappearance of the middle skilled occupations in the last twenty years. They also show that these middle skilled occupation suffered employment losses in the last 2007 recession. Employment losses and unemployment rates, however, are different objects. As the focus of my paper is not about long run trends, but short run fluctuations, I focus here on unemployment rates.

⁵http://usa.ipums.org/usa/volii/census_occtooccsoc.shtml

⁶See the Appendix for details on the definitions and the crosswalk between different databases.

 $^{^7\}mathrm{See}$ http://www.onetcenter.org/content.html for further details.

in my 191 occupations measured in the 2000 census, but the Appendix discusses results with alternative definitions. The education based skill definition is very similar to a measure of non-routine cognitive abilities in the occupations created from the O^*NET data.⁸

2.1 Unemployment fluctuations in occupations

There are two conceptual problems in measuring unemployment rates in occupations. The first is how to assign occupations to the unemployed. From a theoretical point of view at least two approaches are interesting. A backward looking concept assigns the occupation in the last job, and a forward looking concept assigns the occupation in which the unemployed are searching for jobs. Ideally we would like to know both. However, the forward looking concept is not feasible using available datasets, and thus, I use the backward looking concept and assign occupation in the last job to the unemployed.⁹ If workers never changed occupations, the interpretation of this variable is straightforward. In other cases the interpretation is the unemployment rate among those whose most recent affiliation is with a given occupation.^{10,11} The second conceptual problem is that for the new entrant unemployed we cannot assign the occupation at their last job. This paper, thus, only uses unemployment in the experienced labor force, and the new entrant unemployed are excluded.¹²

The two panels of Figure 1 show the unemployment rate in different occupational groups in different time periods. In each occupational group there are 6 bars. The four black ones correspond to years when the aggregate unemployment rate was higher than 5.5 percent, and the gray bars correspond to years when it was lower than 4.5 percent. Panel A classifies the workforce into 10 major groups following the definition of Acemoglu and Autor (2011). There are several interesting patterns in this graph. First, even though the 2007 recession had the largest impact on the labor market, there was nothing unusual in terms of how different occupational groups were affected. Moreover, the last recession was also

 $^{^{8}}$ By using different census years, one gets an almost identical occupational skill variable. As I will show, however, it matters more whether one uses average education or average wage as a proxy of skills in occupations, especially at the lower end of the skill distribution.

⁹In the CPS it is only available for people who worked in the last 5 years.

¹⁰Using long panel datasets such as the PSID, SIPP or HRS, one could create alternative measures. For example, we can fix a reference period; look at the occupational composition of the workforce in this reference period and track the fraction of unemployed over time within these groups. The choice of reference periods, however, would not be straightforward. Another problem is that long panel datasets are usually too small to analyze occupations in detail.

¹¹Another alternative in the CPS is to use the occupation in the longest job last year instead of the last job. By using last year's occupation, however, we seriously underestimate real unemployment in long recessions, since we have to condition on having a job last year. I will show that my main qualitative results do not change if I use this alternative measure, but the level of unemployment shrinks in all occupations.

 $^{^{12}}$ Given that my results are not sensitive to age restrictions, this decision is not restrictive.

comparable in size to the aftermath of the double-dip recession of the 80s. Second, the unemployment rate varies a lot both across occupational groups and over time. While the unemployment rate among managers, professionals and technicians never exceeds five percent, it is rarely below ten percent among operators and laborers. The third interesting finding is that unemployment tends to fluctuate more in occupations where its level is high to begin with, but in a proportional sense the changes are similar. For example, while the unemployment rate among professionals went up from 1.7 percent in 2006-2007 to 3.5 percent in 2009-2012, it went up from 8 percent to 15 percent among operators and laborers. In a proportional sense, we see the doubling of the unemployment rates in almost all occupation groups, but in a level sense the differences are enormous. It is important to note, however, that from a welfare point of view we care about level changes in employment probabilities and, thus, agents care about level, and not percent, changes in their unemployment probabilities.¹³

In terms of the effect of skills, panel A of Figure 1 is rather inconclusive. The level and change of unemployment seem to be the highest among operators, laborers, production workers, food preparation-, cleaning-, personal care and personal services workers. Unfortunately, these categories contain many low- and some middle skilled occupations, and it is hard to draw a conclusion about the effect of skills. The situation is clearer on panel B, where occupations are grouped based on the average years of education in my 191 occupations in the 2000 census. Occupations are grouped so that there are roughly equal number of people in them in the 2000 census, and higher deciles in panel B correspond to better educated workers in the occupation. As opposed to panel A, it is immediately apparent that skills are negatively related to unemployment and to changes in unemployment over time. The effect seems remarkably monotone, with maybe an exception at the lowest skill decile.

The four panels of Figure 6 and 7 in the Appendix show the same results with alternative definitions of occupations and skills. Figure 6 shows that when the occupation last year is used instead of the occupation in the last job, we get a very similar pattern to Figure 1. The only difference is that the level of unemployment is uniformly smaller in the Appendix, because I had to condition on people who had a job last year. As panel B of Figure 7 shows, the results are almost identical if I use the 393 three digit occupations of the 1990 census or if I create occupational skills from the 1990 instead of the 2000 census. As panel A of Figure 7 shows, however, the results are considerably weaker and less

 $^{^{13}}$ Ambiguity averse agents would disprefer employment probability fluctuations even more, because larger fluctuations would make it harder to predict these probabilities.



Figure 1: Unemployment rate by occupational group in different time periods, March CPS 1983-2012 Panel A: In major occupational groups

Panel B: In occupational skill deciles*



*Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census, and higher number means higher skilled occupation. In each occupational group there are 6 bars representing 6 time periods with the ones on the right being the more recent.

monotone when occupational skills are defined in terms of average wage as opposed to average years of education.¹⁴ A usual finding in the job polarization literature is that the middle skilled occupations have been disappearing in the last 20 years. This result appears to be robust to the definition of skills, including average wage, average years of education or task content. Panel A of Figure 7 shows that the business cycle properties of occupations, however, are considerably weaker when wage as opposed to education is used as a proxy of skills. To further elaborate on this issue, Table 6 in the Appendix shows the distribution of major occupational groups within skill deciles based on education (Panel A) and wages (Panel B). We can also see that the skill definition using education follows the major occupation groups better as there are more zero cells (53 as opposed to 41) in panel A compared to panel B. The two skill measures operate similarly at high and middle skilled occupations, but they are quite different at the low skilled ones. The lowest educated occupations are operators, laborers, food and cleaning workers, then production and personal service workers. Recall that in panel A of Figure 1 these were exactly the occupations with the highest unemployment rate and unemployment volatility. However, even though operators, laborers and production workers are uneducated, they are paid comparatively well. From a technical point of view this is the main reason why the skill definition based on education worked better than the one based on wages.

In the next section I will argue that the strong relationship between occupational skills and occupational unemployment patterns should not be though of as the marginal effect of various skills. Instead, it illustrates that occupational skills are positively related to hiring, training and other adjustment costs that induce employers to hoard workers in recessions. Nevertheless, it might be interesting to directly look at various occupational skills from the O*NET project. I use the Abilities section of the O*NET because the dimensions used there are the most similar to the task measures in the Dictionary of Occupational Titles that many researchers used in the job polarization literature.¹⁵ I aggregate abilities into the following 4 categories:¹⁶ 1. Non-routine cognitive abilities (such as reasoning); 2. Routine cognitive abilities (such as attentiveness); 3. Psychomotor abilities (such as finger dexterity)

 $^{^{14}}$ Average wage is computed in the same way as in Acemoglu and Autor (2011): it is the average weakly log wage of the 16-64 year old full-time, full-year workers in the 2000 census. Full-time, full-year workers are those who worked at least 40 weeks last year, and their usual hours of work per week was at least 35. Similarly to Acemoglu and Autor (2011), I excluded observations with weekly earnings less than \$136 in 2008 dollars. The only difference between my definition and that of Acemoglu and Autor (2011) is that I used the 2000 census instead of the 1980 one.

 $^{^{15}}$ The O*NET has many other sections as well, including one on Tasks. From a theoretical point of view, the distinction between tasks and abilities is important. Occupations can be though of as a bundle of tasks, and abilities can be though of as attributes of workers that are important for carrying out these tasks. Task information is useful to test if certain tasks are more important in recession than others. Ability information is useful to test if employers hoard certain types of workers in recessions. In this project I only use abilities, because the 17th edition of O*NET still has incomplete information on occupational tasks.

 $^{^{16}}$ The appendix describes the aggregation, and provides further details about the definitions of my ability measures.

and 4. Physical abilities (such as strength). Table 5 shows correlations between various occupational skill measures in my 191 occupations. The highest correlation is between psychomotor and physical abilities; and between education and non-routine cognitive skills. Figure 8 and 9 show occupational unemployment by occupational skills defined based on the O*NET ability measures. The non-routine cognitive skill content of an occupation gives a very similar picture than my preferred average education based skill measure. Physical skill measures show the reverse picture: the more important physical strength and endurance are in an occupation, the higher is the level and volatility of unemployment. Both of these measures, however, show a weaker association than my simple education based measure, and the other two measures show even weaker associations.

Figure 2 shows the level and log unemployment rates in the ten skill deciles in each March from 1983 to 2012. Qualitatively the evolution of the unemployment rates are similar in each occupation. It starts increasing in recessions, it continues increasing after the recession is over and it starts decreasing only with some lag. Quantitatively, however, these changes are very different. The pattern is, of course, very similar to panel B of Figure 1: both the unemployment rate and the volatility of the unemployment rate are higher in less skilled occupations, but in the log scale (Panel B) we can see that the percentage changes in the unemployment rates over time are similar. What is also fascinating in this figure, is the difference between the rate of recovery of the labor market in the different skill groups. Even though low skilled occupations suffered more during the 2007 recession, the unemployment rate shrank significantly since 2010. At the same time, even though the unemployment rate in middle skilled occupations never reached that of the low skilled occupations, it stuck at its highest level since 2010. This is in line with the findings of Acemoglu and Autor (2011) and Jaimovich and Siu (2012) who showed that middle skilled occupations continued to disappear in the last recession.

Figure 10 in the appendix shows the graphs broken down by gender. The patterns are very similar for both males and females, but the female unemployment rate is somewhat smaller in level and volatility especially prior to the 2007 recession. In the 2007 recession the male and female unemployment rate followed a very similar pattern.

To further illustrate the power of my education based occupational skill variable, I run OLS regressions of unemployment probabilities on skills, different demographic variables and their interaction with GDP growth. GDP growth is defined as the log difference between GDP in the previous calendar year and

.18 .16 Unemployment rate .14 .12 .1 .08 .06 .04 .02 0 2010 2013 1983 1986 1989 1992 1995 1998 2001 2004 2007 Year 5th 1st 2nd 3rd 4th 6th 7th 8th 9th 10th -



Panel B: Log unemployment rates



*Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census, and higher number means higher skilled occupation. The gray areas indicate NBER recession dates.

the year before that.¹⁷ Table 1 shows the output of the regressions together with an estimate of the marginal effect of a 10% higher GDP growth on unemployment at various percentiles of the skill or education distribution. By comparing columns 1 and 2 we can see that occupational skill is a better predictor of unemployment than education, which is remarkable since my occupational skill variable is defined as an occupation specific average years of education. A 10% higher GDP growth induces 7.3 percentage point decrease in unemployment in occupations at the 5th percentile of the skill distribution, and a 7.4 percentage point decrease at the 5th percentile of the education distribution. At the 95th percentiles, however, these numbers are 1.5 and 3 percentage points, respectively, which means that by going from the 95th to the 5th percentiles, the fluctuation in unemployment increases 4.9 times when occupational skills, but only 2.4 times when education is considered. By comparing columns 3 and 4 we can see further evidence that occupational skills are better predictors of unemployment: more than half of the educational difference in unemployment can be explained by my occupational skill variable, but at the same time, occupational skill does not decrease much after controlling for the actual level of education of workers. We can also see, however, that education and other demographic variables all remain significant, which means that within occupational skill groups there is still systematic differences in the unemployment prospects of workers. This might indicate that I used too broad occupational groups, or that these variables actually predict unemployment.

Table 7 in the appendix shows the same results when a log-specification is used. The model fits the following regression with non-linear least squares:

$$U_{it} = \exp\left(X_{it}\beta\right) + u_{it}$$

and the marginal effects of GDP growth at various percentiles of the skill and education distribution are evaluated at the mean GDP growth.¹⁸ The estimated models are non-linear and thus it is not easy to read the output of these models. Nevertheless, the implied effects of GDP growth on various percentiles of the skill and education distribution are very close to the ones based on the level model in Table 1. Therefore I will use the level model from now on.¹⁹

Table 8 in the appendix repeats this exercise in different age, gender and race groups. The occupational

 $^{^{17}}$ Other lags of GDP has further explanatory power, but I only use this single variable to make the table easier to read and because this GDP variable has the strongest load on unemployment.

 $^{^{18}\}mathrm{See}$ the appendix for further details.

¹⁹The R-squared values in the non-linear models are significantly higher than in the level models, which might indicate that the log specification gives a better fit. Nevertheless, because the level model implies similar estimates and it is much easier to work with, I use the level model, instead.

distribution is kept the same in all regressions. As we can see, the gender difference in the level and volatility of unemployment remains large even when occupational skills are controlled. At the 5th percentile of the skill distribution, a 10% increase in GDP causes a 9.2 percentage point increase in the male unemployment rate, while it leads to only 4.5 percentage point increase among females. The corresponding numbers at the 95th percentile are 1.1 percentage point (males) and 2.4 percentage point (females). Thus, male unemployment rate is more volatile at the bottom of the skill distribution and less volatile at the top of it. Race differentials in the level and volatility of unemployment are less stark, when occupational skills are controlled. Even more interesting is the differences among age groups. The level of unemployment is lower for older people, but this difference is very similar at all occupational skill levels.

Table 9 in the appendix shows the same regressions by using alternative occupational skill measures from either the census (average education or wage), or the O^*NET . As we can see my education based skill measure is the strongest predictor of occupational unemployment. Non-routine cognitive skills behave similarly to the education based measure, and psychomotor and physical abilities work in the opposite way.²⁰

Finally I show scatter plots of unemployment rates in each of the 191 occupation I defined above. The two panels of Figure 11 show the unemployment rates in a period of favorable labor market conditions (2006-2007) and in a period of weak labor market (2009-2012). Figure 3 shows the change in the unemployment rates between these two periods. In all figures I colored the occupations based on the major occupation groups they belong to. The general pattern is very clear: lower skilled occupations have higher unemployment in all periods, and they are more strongly hit by the business cycle. It seems to be true both within and across major occupation groups. Moreover, we can also see that there is an enormous heterogeneity at the low end of the skill distribution that is not explained by my skill proxy.

2.2 Job-loss and job-finding probabilities in occupations

There is a debate in the macro labor literature about the relative importance of job-separation and jobfinding probabilities in explaining fluctuations in the aggregate unemployment rate. The usual finding

 $^{^{20}}$ When one runs a horse race between the alternative measures, the education based one comes down to be the strongest, but all the others remain significant with the expected sign, too.

Table 1: OLS regressi	ions of unen	nployment,	March CPS,	1983-2012
	[1]	[2]	[3]	[4]
dln(GDP), DG	-0.5	-1.236	-1.459	-0.964
	$[0.007]^{**}$	$[0.037]^{**}$	$[0.046]^{**}$	$[0.056]^{**}$
Occupational skill, SE	-0.036			-0.025
	$[0.000]^{**}$			$[0.000]^{**}$
SE X DG	0.176			0.135
	$[0.007]^{**}$			$[0.009]^{**}$
Years of education, E		-0.011	-0.01	-0.004
		$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$
E X DG		0.055	0.05	0.016
		$[0.003]^{**}$	$[0.003]^{**}$	$[0.004]^{**}$
Age, A			-0.001	-0.001
			$[0.000]^{**}$	$[0.000]^{**}$
A X DG			0.002	0.002
			$[0.001]^{**}$	$[0.001]^{**}$
Female, F			-0.018	-0.014
			$[0.000]^{**}$	$[0.000]^{**}$
F X DG			0.25	0.224
			$[0.015]^{**}$	$[0.015]^{**}$
White, W			-0.037	-0.034
			$[0.001]^{**}$	$[0.001]^{**}$
W X DG			0.082	0.074
			$[0.020]^{**}$	$[0.020]^{**}$
Constant	0.074	0.222	0.305	0.216
	$[0.000]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.002]^{**}$
N	2462447	2462447	2462447	2462447
R squared	0.02	0.014	0.022	0.027
Marginal effect of 10% higher	GDP grow	th on unem	ployment at	various percentiles
of the skill distribution (mode	el 1) and th	e education	distribution	$(model \ 2)$
5th percentile	-0.073	-0.074		
	$[0.001]^{**}$	$[0.001]^{**}$		
25th percentile	-0.063	-0.058		
	$[0.001]^{**}$	$[0.001]^{**}$		
Median	-0.05	-0.052		
	$[0.001]^{**}$	$[0.001]^{**}$		
75th percentile	-0.037	-0.036		
	$[0.001]^{**}$	$[0.001]^{**}$		
95th percentile	-0.015	-0.03		
	$[0.002]^{**}$	$[0.001]^{**}$		
Ratio: 5th / 95th percentile	4.941	2.442		
	$[0.579]^{**}$	$[0.128]^{**}$		

*dln(GDP) is defined as the log difference between GDP in the previous calendar year and the year before that; Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations standardized to have zero mean and standard deviation 1 in the 2000 census.

Figure 3: Average increase in unemployment rate by occupations between 2006-2007 and 2009-2012, March CPS, 2006-2012, colored by 10 major groups



is that the job-finding probability is relatively more important, but the job-loss probability also has explanatory power, especially at the early phases of recessions. (Shimer, 2012; Davis et al., 2006; Elsby et al., 2009; Fujita and Ramey, 2009). In this section I derive job-loss and job-finding probabilities by occupations. Elsby et al. (2010) found that the job-finding probability fluctuates remarkably similarly in different demographic groups, while the job-loss probability varies greatly. I will show that this holds for occupations as well. Even though the job-finding probabilities are very volatile, they fluctuate just as much in skilled occupations as in unskilled ones. The job-loss probabilities, however, are smaller and more smooth in skilled occupations compared to unskilled ones.

For this exercise I merge the monthly CPS from September 1995 to January 2013²¹. The job-loss probability is defined as the ratio of people in an occupation who become unemployed by the next month. The job-finding probability is defined as the ratio of people who find a job by the next month among those whose last job was in a given occupation. Occupation switches are not modeled here. I

 $^{^{21}}$ Going back further in time is possible, but tricky. CPS changed the individual identifiers a couple of times in the past and at these dates we cannot merge consecutive CPS waves. The last change was in the entire summer of 1995, when for three months the identifiers were changing every month. Before 1994, the occupation data is less reliable than after 1994. In January 1994 CPS introduced CAPI, they preloaded the previous occupations of workers and only asked about changes (or potential miscoding) in occupations. Thus, the pre-1994 occupation data contains considerably more measurement error. For occupational unemployment, it might not be a big problem, but in later sections I will also analyze occupation switches for which the pre-1994 values are much less reliable.

only estimate whether someone in a given occupation is employed or not employed by the next month.

Figure 4 and 12 show how the job-loss $(E \to U)$ and job-finding $(U \to E)$ probabilities evolved between 1996 and 2013 in different occupations. As we can see the job-finding probabilities are considerably more volatile over the business cycle than the job-loss probabilities, which is in line with the macro labor literature cited above. For example, the average monthly job-finding probability was over 30 percent in 1998, while it was below 20 percent in 2009. However, even if the job-finding probabilities are more volatile, we see little difference between occupational groups. It seems that it is not significantly easier for the skilled to find a job neither in booms nor in busts. The sample average of the job-finding probability in the total 1996-2013 period was the lowest for managers (24 percent) and the highest for professional workers (29 percent). This difference is not huge, and it does not seem to systematically vary by skills.

The job-loss probabilities, however, are smoother and they basically track the pattern of occupational unemployment shown in Figure 1 and 2. The job-loss probabilities are very small and they do not vary much in the most skilled occupations (like managers, professional workers, and technicians) but they are high and volatile for less skilled occupations. It seems, thus, that even though separations are less important to understand aggregate fluctuations in unemployment, they are very important to understand the differences between occupational unemployment.

Table 2 shows the regression versions of Figure 4. In the regressions I use various lags of quarterly GDP growth data in the following way:

$$P_{ot} = \beta_0 + \beta_1 S_o + (1 + \beta_2 S_o) \left(\sum_{l=1}^{L} \gamma_l \left(\ln GDP_{t-l} - \ln GDP_{t-l-1} \right) \right) + u_{ot}$$
(1)

where o indexes occupations. The specification in (1) restricts the shape of the impulse response function to be the same in every occupations, and β_2 determines the scale. A negative β_2 implies a smoother impulse response in skilled occupations. The transition probabilities are seasonally adjusted in each occupation separately²² and the regressions are weighted by the size of the occupation-month cells.²³ As we can see in Table 2, GDP growth has a sharp, but short-lasting negative effect on job-loss

 $^{^{22}}$ In each occupation, I regressed the transition probabilities on month dummies and I added up the residuals and the occupational means.

 $^{^{23}}$ The weight is is the sum of the employed and unemployed persons in each occupation in every months.





Panel A: job-loss probabilities

Panel B: job-finding probabilities



*The figure shows yearly moving averages of monthly job-loss and job-finding probabilities using the monthly CPS. For the unemployed, the occupation at the last job is used. Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census, and higher number means higher skilled occupation. The gray areas indicate NBER recession dates.

probabilities, but the response of the job-finding probabilities is more persistent. This is in line with the macro labor literature of aggregate job-loss and job-finding probabilities cited in the introduction. Even more important is that while there are large occupational differences in the level and volatility of job-loss probabilities, there are no significant differences in the level and volatility of job-finding probabilities. Differences in occupational unemployment, thus, are only driven by differences in the incidence of job-loss.

3 Possible economic explanations

In this section I collect six alternative hypotheses that might explain the large heterogeneity in occupational unemployment and provide simple tests of them. In the end I find that differential labor hoarding by occupational skills is the only explanation that is consistent with all tests. First, let me briefly outline the six hypotheses.

- 1. Industrial composition: Low-skilled occupations might be clustered in industries that are disproportionally more strongly hit by recessions, such as durable good manufacturing.
- 2. Quality adjustment (à la Reder): The skilled might crowd out the unskilled from the labor market in recessions as employers prefer hiring skilled workers even into low skilled jobs.
- 3. Productivity differences: Some skills might be useful in recessions and they might be positively correlated with the general skill proxies that I use in this project.
- 4. Employment flexibility: Skilled workers might have more flexible work arrangements with their employers that make work adjustments more likely on the intensive as opposed to the extensive margin.
- 5. Labor hoarding due to adjustment costs: Hiring, training and other adjustment costs might be higher in skilled occupations. Due to these costs firms might decide to hoard their skilled workers to save on their future rehiring costs.
- 6. Labor hoarding due to firm specific skills: Workers in skilled occupations might acquire more firm specific skills which are valued by the employer.

	Job	-loss	Job-f	inding
	[1]	[2]	[3]	[4]
Occupational skill	-0.007	-0.007	0.003	0.003
-	$[0.000]^{**}$	[0.000]**	$[0.001]^*$	[0.001]
Interaction	-0.401	-0.402	-0.010	0.007
	$[0.038]^{**}$	$[0.035]^{**}$	[0.025]	[0.020]
dln(GDP), lag 1	-0.084	-0.075	1.394	1.075
	$[0.009]^{**}$	$[0.009]^{**}$	$[0.156]^{**}$	$[0.155]^{**}$
dln(GDP), lag 2	-0.061	-0.070	1.076	1.450
	$[0.009]^{**}$	$[0.010]^{**}$	$[0.167]^{**}$	$[0.167]^{**}$
dln(GDP), lag 3	-0.056	-0.043	1.596	1.064
	$[0.010]^{**}$	$[0.010]^{**}$	$[0.167]^{**}$	$[0.170]^{**}$
dln(GDP), lag 4	-0.077	-0.050	2.742	1.361
	$[0.009]^{**}$	$[0.010]^{**}$	$[0.154]^{**}$	$[0.169]^{**}$
dln(GDP), lag 5		-0.022		0.823
		$[0.010]^*$		$[0.168]^{**}$
dln(GDP), lag 6		-0.021		1.504
		$[0.009]^*$		$[0.164]^{**}$
dln(GDP), lag 7		-0.065		2.751
		$[0.009]^{**}$		$[0.152]^{**}$
Constant	0.015	0.016	0.223	0.204
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$
Ν	39919	39919	36900	36900
R squared	0.286	0.288	0.040	0.063
Marginal effect of 1% higher	GDP growt	h in each qu	uarter on un	employment
at various percentiles of the s	skill distribu	tion		
5th percentile	-0.004	-0.005	0.069	0.099
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.003]^{**}$	$[0.003]^{**}$
25th percentile	-0.004	-0.004	0.069	0.100
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.002]^{**}$	$[0.003]^{**}$
Median	-0.003	-0.004	0.068	0.100
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$
75th percentile	-0.002	-0.003	0.068	0.101
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$
95th percentile	-0.000	-0.001	0.067	0.102
	$[0.000]^*$	$[0.000]^*$	$[0.004]^{**}$	$[0.004]^{**}$
Ratio: 5th / 95th percentile	9.465	9.540	1.036	0.977
	[4.938]	$[4.641]^*$	$[0.089]^{**}$	$[0.065]^{**}$

Table 2: Non-linear least squares models of job-loss and job-finding probabilities, Monthly CPS, from September 1995 to January 2013

[4.938] [4.641]* $[0.089]^{**}$ $[0.065]^{**}$ *The left hand side contains occupation specific job-loss and job-finding probabilities by months. The probabilities are seasonally adjusted separately for each occupation, by collecting the residuals from a regression of occupation specific probabilities on month dummies. The regressions are weighted by the number of employed and unemployed workers in a given month in a given occupation. The GDP growth is computed from aggregate quarterly data. For the unemployed, the occupation at the last job is used. Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census, and higher number means higher skilled occupation. The interaction term means β_2 in specification (1).

3.1 Industrial composition

Different industries use different occupations, and thus, it is possible that the differential effect of the business cycle on occupations is only a consequence of industry-specific shocks. For example, a common belief is that manufacturing uses more unskilled production workers, and this industry suffers relatively more during contractions.

Even though the industry composition hypothesis could explain large differences in occupational unemployment, it cannot explain why the job-finding probabilities are so similar in different occupations. In fact the hypothesis predicts that occupations with relatively stable unemployment profiles should have relatively stable job-finding probabilities, too. However, this is not what we see in the data.

In order to test this hypothesis more formally, I created industry specific GDP series using the BEA reports. I had to make some aggregation of the industries to get a comparable specification with the CPS.²⁴ I ended up with 38 industries. Figure 13 in the appendix shows that industry specific GDP growth is only weakly related to occupational skills in these industries. We can see that in the most skilled industries (professional services and public administration) the volatility of GDP growth is small, and the GDP in 2007-2009 was not strongly hit, either. However, at lower skilled industries the relationship is quite weak. Thus, it is unlikely that the industry composition hypothesis is responsible for the occupational unemployment patterns.

A more formal test of the hypothesis can be carried out by comparing the β_2 coefficients in the following regressions:

$$U_i = \beta_0 + \beta_1 S_i + (1 + \beta_2 S_i) \left[\beta_3 d \ln \left(GDP_t\right)\right] + \varepsilon_i \tag{2}$$

$$U_i = \beta_0 + \beta_1 S_i + (1 + \beta_2 S_i) \left[\beta_3 d \ln \left(GDP_t\right) + \beta_4 d \ln \left(GDP_{jt}\right)\right] + \varepsilon_i, \tag{3}$$

where $d \ln (GDP_{jt})$ denotes industry GDP growth. The industry composition hypothesis predicts a large negative β_2 in equation (2) and a close to zero β_2 in equation (3). Table 3 shows, however, that the interaction term β_2 actually decreases more after industry specific shocks are taken into account.

 $^{^{24}}$ In fact the crosswalk was even more complicated because the CPS definitions also changed over time. A major change occurred in 2003 where the previously used Standard Industrial Classification (SIC) got replaced by the North American Industry Classification System (NAICS). The BEA information is also based on NAICS and thus the crosswalk is simpler for the post 2003 years. There has also been two minor changes in the industry classification in CPS in 1992 and 2009.

	[1]	[2]
Occupational skill	-0.036	-0.036
	$[0.000]^{**}$	$[0.000]^{**}$
Interaction	-0.352	-0.410
	$[0.015]^{**}$	$[0.016]^{**}$
dln(GDP)	-0.500	-0.391
	$[0.007]^{**}$	$[0.008]^{**}$
dln(Industry GDP)		-0.080
		[0.003]**
Constant	0.074	0.074
	$[0.000]^{**}$	$[0.000]^{**}$
N	2462447	2462447
R squared	0.020	0.020
Marginal effect of 10% highe	r GDP grow	th on unemployment
at various percentiles of the s	skill distribu	tion
5th percentile	-0.073	-0.072
	[0.001]**	[0.001]**
25th percentile	-0.063	-0.061
	$[0.001]^{**}$	$[0.001]^{**}$
Median	-0.050	-0.047
	[0.001]**	[0.001]**
75th percentile	-0.037	-0.032
	[0.001]**	[0.001]**
95th percentile	-0.015	-0.008
	$[0.002]^{**}$	[0.002]**
Ratio: 5th / 95th percentile	4.941	8.563
	$[0.578]^{**}$	$[1.666]^{**}$

Table 3: Testing the industry composition hypothesis with industry GDP data, March CPS 1983-2012

*dln(GDP) and dln(Industry GDP) are defined as the log difference between aggregate and industry GDP in the previous calendar year and the year before that; Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations standardized to have zero mean and standard deviation 1 in the 2000 census. For the unemployed, the occupation and industry at the last job is used. The interaction term means β_2 in specification (2) and (3). The percentiles at model 2 show the marginal effect of a 10% higher GDP growth both in the aggregate and in the industry of the worker.

By comparing the effects at various percentiles of the skill distribution we can see that this difference is driven by the highest skilled occupations.

Overall, the industry composition hypothesis cannot explain the large differences in occupational unemployment patterns.

3.2 Quality adjustment

It is possible that people in skilled occupations can get jobs in both skilled and unskilled ones and in recessions they crowd out the unskilled from the labor market. This could happen if skilled workers increase their search effort for unskilled jobs (Albrecht and Vroman, 2002; Dolado et al., 2009; Khalifa, 2009; Chassamboulli, 2011) or if firms raise hiring standards in recessions (Reder, 1955; Teulings, 1993; van Ours and Ridder, 1995; Evans, 1999; Devereux, 2004; Büttner et al., 2010). This mechanism can explain why the unemployment rate is more volatile in unskilled occupations.

The mechanism also predicts that the job-finding probability fluctuates more for the unskilled. As we saw in the previous chapter, however, the job-finding probability follows a broadly similar pattern in all occupations. Thus, the quality adjustment hypothesis is very unlikely to explain why different occupations are affected by the business cycle so differently.

Nevertheless I also tested this hypothesis directly. The hypothesis predicts that in recessions we see an increase in flows from high to low skilled jobs, while in booms the trend reverses. Panel A of Figure 5 shows the average education level of job-switchers in different occupational skill deciles in the last 30 years in the March CPS.²⁵ The hypothesis predicts then in recessions the average education levels converge to each other, and in booms they diverge. We do not see that in the data. Panel B shows the average occupational skill in the next job among those who changed occupations between two months. Again, there is not evidence for any convergence in busts and divergence in booms. Overall the quality adjustment hypothesis cannot explain the large differences in the occupational unemployment patterns.

3.3 Productivity differences

Some skills might be useful in recessions and they might be positively correlated with the general skill proxies that I use in this paper. For example, even if a manufacturing firm wants to lay off some production workers, it might still need as many (or even more) market analysts. Recessions, thus, might affect occupations differently just because some tasks carried out in some occupations are relatively more useful in recessions than tasks carried out in other occupations.

The productivity hypothesis, similarly to the previous quality adjustment one, also predicts that the job-finding probability fluctuates less for skilled workers. As long as skilled workers are more useful

 $^{^{25}}$ Job-switchers are those whose occupation last year was different from the occupation in the March CPS.





Panel B: Monthly CPS, 1 year moving average



*Panel A shows the average years of education measured in the CPS in 10 occupational skill categories among those whose occupation last year was not the same as their occupation in the March CPS. Panel B shows 1 year moving averages of the average occupational skill in the next job among those who switched occupations between two months. Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. The light gray areas indicate NBER recession dates. The dark gray column on Panel A at 1992 indicates a change in the education measurement in the CPS.

in recessions, the job-finding probabilities of the skilled should be less volatile than the ones of the unskilled. As we do not see any evidence for this in the data it seems very unlikely that this mechanism plays an important role in understanding differences in the unemployment patterns of occupations.

3.4 Employment flexibility

In principle employers can adjust labor both on the extensive and the intensive margins. If, for some reason, it is easier to adjust skilled labor on the intensive margin, we would expect smaller fluctuation in unemployment among skilled workers. To test this hypothesis, I computed average hours of work per week by occupations over time among those who worked positive hours. Table 4 shows, however, that people actually work longer hours in skilled occupations, and their work hours are affected less by the business cycle. Figure 14 in the appendix shows the same results in various occupation groups. We can see that average hours worked per week is procyclical, and lower skilled occupations are somewhat more affected on the intensive margin. Altogether, work flexibility is not responsible for seeing less unemployment fluctuations in skilled occupations.

3.5 Labor hoarding due to adjustment costs

The available empirical evidence suggest that hiring, training and other adjustment costs are higher in skilled occupations even proportionally to the wage rate (Manning, 2011). These costs might induce firms to hoard skilled workers in order to save on their rehiring costs in the future. At the same time, unskilled workers are less likely to be hoarded because it is comparatively cheaper to lay them off in recessions and rehire them in booms. Oi (1962) found that recruitment costs are lower and labor turnover is higher among "Common laborers". Since Oi's work many papers provided evidence for labor hoarding, but I am not aware of any papers that analyzed occupational differences in the extent of labor hoarding.

This mechanism can explain the empirical finding in the previous section that the job-finding probabilities are so similar while unemployment rates are so different across occupations. In recessions employers have excess labor in each occupation on average. Consequently, the rate of hiring is cut back, and the job-finding rate decreases. The fact that the job-finding probabilities empirically move

	[1]	[2]
dln(GDP), DG	14.947	21.54
	$[0.448]^{**}$	$[3.321]^{**}$
Occupational skill, SE	2.015	1.432
	$[0.014]^{**}$	$[0.018]^{**}$
SE X DG	-1.85	-2.67
	$[0.437]^{**}$	$[0.545]^{**}$
Years of education, E		0.359
		[0.007]**
E X DG		0.743
		$[0.206]^{**}$
Age, A		0.163
		$[0.001]^{**}$
A X DG		-0.113
		[0.035]**
Female, F		-6.222
		$[0.028]^{**}$
F X DG		-13.951
		$[0.861]^{**}$
White, W		0.043
		[0.038]
W X DG		-1.681
		[1.196]
Constant	38.561	30.363
	$[0.015]^{**}$	$[0.108]^{**}$
Ν	2231454	2231454
R squared	0.022	0.106
Marginal effect of 10% higher	GDP growt	h on hours worked
per week at various percentil	es of the skill	distribution
5th percentile	1.728	2.49
	$[0.073]^{**}$	$[0.297]^{**}$
25th percentile	1.62	2.335
	$[0.055]^{**}$	$[0.312]^{**}$
Median	1.493	2.152
	$[0.045]^{**}$	$[0.332]^{**}$
75th percentile	1.355	1.952
	$[0.054]^{**}$	$[0.358]^{**}$
95th percentile	1.125	1.62
	$[0.095]^{**}$	$[0.405]^{**}$
Ratio: 5th / 95th percentile	1.536	1.537
	$ 0.168 ^{**}$	$ 0.228 ^{**}$

Table 4: OLS on average hours of work last week by occupational groups*, March CPS 1983-2012

^{*}dln(GDP) is defined as the log difference between GDP in the previous calendar year and the year before that; Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations standardized to have zero mean and standard deviation 1 in the 2000 census.

together indicates that skilled occupations are just as useful (or useless) in recessions as unskilled occupations. However, given that adjustment costs are higher in skilled occupations, employers are more likely to hoard their skilled employees and lay off their unskilled ones. If adjustment costs are monotonically increasing with skill, we expect unemployment volatility to monotonically decrease with the skill level of the occupation. It would be very useful to have a direct measure of adjustment costs by occupations so that this hypothesis can be tested directly.

3.6 Labor hoarding due to firm specific skills

Workers in skilled occupations might acquire more firm specific skills than workers in unskilled occupations. The firm specific skill is valued by both the worker and the employee and thus they might decide to keep the match even in recessionary times when the match does not produce much value. This hypothesis is closely related to the previous one as the presence of firm specific skills might be a reason for hoarding labor.

Labor hoarding models, however, usually emphasize adjustment costs such as recruitment, training or lay-off costs. These costs appear at the beginning and at the end of the employment spells. Firm specific skills, however, perhaps increase with tenure more gradually. It is interesting, thus, to test whether occupational differences in layoff probabilities change with tenure, or not. Imagine, for example, that we find that among newly hired workers the difference in the lay-off probabilities among occupations is small, but this differential is increasing with tenure. This would be indirect evidence that firm specific skills, as opposed to adjustment costs are responsible for hoarding skilled workers.

Unfortunately, there is no information about tenure in the CPS. A second best approach is to use age. Younger workers, on average, have shorter employment spells and less firm specific human capital for at least two reasons. First, they have spent less time on the labor market, and thus, they had less chance to acquire skills. Second, younger workers tend to switch employers more often than older workers.

In Table 8 we have already seen that older workers are less likely to be laid-off in recessions, which indicates that employers do value labor market experience when they decide about layoffs. However, we also saw that the occupational differences in unemployment are just as large among the young as they are among the old. If employers hoard their skilled workers because skilled workers acquire more firm specific human capital, we should see smaller occupational unemployment differences among the young. Based on Table 8 we can conclude that if firm specific skills are responsible for the large occupational unemployment differences, then these firm specific skills are acquired early on, and thus, they are more similar to adjustment costs.

4 A simple model of occupation specific labor hoarding

The model I lay out here intends to illustrate that under rigid wages and adjustment costs there is a possibility for a market failure in certain low-skilled occupations. When wages are rigid and adjustment costs are positive, lay-offs are never socially optimal. Instead, workers are hoarded in recessions and wages are equal to the average of the worker's productivity minus the adjustment costs over the business cycle. When adjustment costs are high enough, the market outcome is the same as the social optimum. However, when adjustment costs are low, the market outcome features inefficient lay-offs, because firms cannot commit to future employment in exchange for some reduction in wages. Even though workers would prefer lower wages for assured future employment, such contracts are not credible.

I am not aware of a model that analyzed the social efficiency of labor hoarding.²⁶ Labor hoarding models in macroeconomics are usually interested in firms' optimal adjustment of labor and the resulting volatility and persistence of aggregate unemployment. The implicit contract literature tries to explain why wages are rigid and layoffs are frequent if firms and workers can write contracts that are contingent on future economic conditions and workers are risk-averse. Search and matching models usually assume that the Hosios condition holds, which implies efficiency, although there has been some work analyzing more general models.

4.1 The setup of the model

1. In each period there is a recession (R) with probability p^R and a boom (B) with probability $p^B \equiv 1 - p^R$. These states are uncorrelated over time.

 $^{^{26}}$ I am in the process of finding relevant labor hoarding or related models that can be used to analyze occupational differences in unemployment patterns; and to analyze policy implications for influencing adjustment costs. Any suggestions would be greatly appreciated.

- 2. There are O occupations indexed by $o \in \{1, ..., O\}$. Firms either use zero or one workers and their problem is to optimally hire and lay-off over the business cycle. Firms can offer two types of contract, but they cannot commit to them:
 - (a) A stable contract S: the firm offers to keep its worker in recessions and if it does not have a worker, it hires one in booms.
 - (b) An unstable contract U: the firm fires its worker in recessions and if it does not have a worker, it hires one in booms.
- 3. Employers can hire someone in occupation o after paying a fixed hiring cost, $h_o \ge 0$, they can fire workers after paying a fix firing cost $f_o \ge 0$, and they discount the future with discount rate β .
- 4. Wages (w_o) are rigid in each occupation and are determined in equilibrium. Free entry among firms assures that the value of a new hire is zero.
- 5. Workers are homogenous within occupations and they cannot change their occupations. Their productivity depends on the occupation and the state of the economy

$$p_o = a_t s_o$$

with $a_t \in \{a^R; a^B\} \ a^B > a^R > 0.$

6. Employees quit a job with exogenous probability γ_o , and they maximize their expected wages, which is the product of the rigid wage and employment probability, $W_o = w_o \Pr(E_o)$. The outside option of workers is 0.

The model asks two questions: Which contract is socially optimal and under what conditions does the market achieve it? In the appendix I show that the equilibrium wage in the stable and the unstable contracts are:

$$w_{o}^{S} = a^{B}s_{o} - \beta p^{R} (1 - \gamma_{o}) (a^{B} - a^{R}) s_{o} - h_{o} (1 - \beta (1 - \gamma_{o}))$$
$$w_{o}^{U} = a^{B}s_{o} - \beta p^{R} (1 - \gamma_{o}) f_{o} - h_{o} (1 - \beta p^{B} (1 - \gamma_{o}))$$

In both contracts, wages are positively related to the productivity terms $(a^B, a^R \text{ and } s^o)$, to the probability of a boom (p^B) and negatively related to the adjustment costs $(h_o \text{ and } f_o)$. The appendix also shows that the expected wages of workers are

$$W_{o}^{S} = \frac{p^{B}}{p^{B} + p^{R}\gamma_{o}} \left[a^{B}s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) \left(a^{B} - a^{R} \right) s_{o} - h_{o} \left(1 - \beta \left(1 - \gamma_{o} \right) \right) \right]$$

$$W_{o}^{U} = p^{B} \left(a^{B}s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) f_{o} - h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o} \right) \right) \right)$$

4.2 The social optimum

In this model firms make zero profit, and thus, the socially optimal contract maximizes the expected wage of workers. In the appendix I show that the unstable contract is never socially optimal: If the expected wage of the unstable contract is positive, then the stable contract always offers a higher expected wage:

$$W_o^U > 0$$

 \downarrow
 $W_o^S > W_o^U$

Hoarding labor socially dominates lay-offs, because it minimizes the number of instances when adjustment costs are paid. If firms can commit to future employment, that is, they can credibly offer the stable contract, then the market always achieves the social optimum. However, if firms cannot commit to future employment, the market outcome might be sub-optimal.

4.3 The market outcome

If firms cannot commit to future employment, then workers always pick a contract that maximizes perperiod wages. This is the case, because firms cannot credibly assure future employment in exchange for a reduction in per-period wages. In the appendix I show that the the socially optimal stable contract is achieved in occupations where

$$w_o^S \geq w_o^U$$

$$\overset{\Psi}{h_o + f_o} \geq (a^B - a^R) s_o$$

$$(4)$$

This is quite intuitive. The left hand side of (4) is the marginal benefit of keeping workers in recessions (saving adjustment costs) and the right hand side is the marginal cost (the reduction in productivity). Thus, in occupations where the hiring and firing costs are large enough, firms can credibly offer the socially optimal stable contract. Based on the empirical hiring cost literature these occupations are typically the skilled ones, because recruitment and training costs are proportionally higher in these jobs. In unskilled occupations, however, firms cannot credibly offer the socially optimal stable contract, and we expect inefficiently large fluctuations in unemployment.

≙

4.4 Discussion of the model

Even though this simple model cannot be used to calibrate a realistic labor market and analyze welfare implications of alternative policies, it does have some interesting qualitative policy suggestions. The first is that supporting job-creation in recessions might not be a good idea, since labor hoarding is pervasive in the economy and firms have excess labor in recessions anyways. The second implication is that the large unemployment fluctuations in many unskilled jobs might indicate a market failure that might be mitigated with appropriate policies. The market failure is the consequence of the relatively rigid wages and the relatively low adjustment costs in low-skilled jobs. A good policy, thus, should either aim at making wages less rigid or increasing the adjustment costs. Out of the alternatives, influencing firing costs seems to be the easiest to implement,²⁷ and by appropriately targeting the policy we might be able to maximize the social benefit and minimize the cost. One idea is to increase firing costs only in recessions and only in occupations where adjustment costs are too low. This way the labor market can clear in normal economic circumstances, and the policy targets the population that is most affected by the market failure.

²⁷Wage rigidity might be influenced by generating inflation, but that, of course, creates many additional problems.

5 Conclusion

This paper documented large differences in unemployment rates and unemployment fluctuations across occupations. By looking at detailed occupations I found that the skill level of an occupations had a strong negative effect on unemployment prospects in all recessions of the last 30 years. This was robust to different measures of general skills, but education-based measures led to the strongest and most monotone results. I decomposed unemployment fluctuations into changes in the job-loss and the job-finding probabilities in each occupation. I found that even though separations are less important for aggregate fluctuations in unemployment, they are more important for understanding the differences between occupational unemployment, as the job-finding probability followed a highly similar pattern across occupations. I argued that this result is in line with a model in which employers are more likely to hoard skilled workers due to higher recruitment/training and other adjustment costs in skilled jobs. I also considered five other hypotheses that can explain why unemployment fluctuations are lower in skilled occupations. These were the industrial composition; the quality adjustment; the productivity differences; the employment flexibility and the firm specific skills hypotheses. I provided simple tests of these alternative hypotheses and I found that they were considerably less important to understand occupational differences in unemployment patterns than the labor hoarding hypothesis.

Future research will compare the evolution of job-loss and job-finding probabilities in the US to those in Germany. Even though the more flexible labor laws of the US could be better than the more protective German ones in normal economic circumstances, it is possible that at least in unskilled occupations, unemployment can grow faster in the US in recessions. I plan to propose a new labor protection system which maximizes the welfare of citizens by concentrating protection in risky occupations and uncertain times and therefore minimizing the incidence of unemployment and decreasing the burden on the unemployment insurance systems.

References

- Abraham, K. G. and J. R. Spletzer (2009, May). New evidence on the returns to job skills. American Economic Review 99(2), 52–57.
- Acemoglu, D. (1999, December). Changes in unemployment and wage inequality: An alternative theory and some evidence. *American Economic Review* 89(5), 1259–1278.

- Acemoglu, D. and D. Autor (2011). Skills, Tasks and Technologies: Implications for Employment and Earnings, Volume 4 of Handbook of Labor Economics, Chapter 12, pp. 1043–1171. Elsevier.
- Albrecht, J. and S. Vroman (2002, February). A matching model with endogenous skill requirements. International Economic Review 43(1), 283–305.
- Autor, D. H., L. F. Katz, and M. S. Kearney (2006, May). The polarization of the u.s. labor market. American Economic Review 96(2), 189–194.
- Autor, D. H., F. Levy, and R. J. Murnane (2003, November). The skill content of recent technological change: An empirical exploration. *The Quarterly Journal of Economics* 118(4), 1279–1333.
- Azariadis, C. (1975, December). Implicit contracts and underemployment equilibria. Journal of Political Economy 83(6), 1183–1202.
- Azariadis, C. (1976, February). On the incidence of unemployment. Review of Economic Studies 43(1), 115–25.
- Baily, M. N. (1974, January). Wages and employment under uncertain demand. Review of Economic Studies 41(1), 37–50.
- Basu, S. and M. S. Kimball (1997, February). Cyclical productivity with unobserved input variation. NBER Working Papers 5915, National Bureau of Economic Research, Inc.
- Blatter, M., S. Muehlemann, and S. Schenker (2012). The costs of hiring skilled workers. European Economic Review 56(1), 20 – 35.
- Brown, C. (1990). Empirical evidence on private training. In *Research in Labor Economics*, Volume 11, pp. 97–113. JAI Press Inc.
- Büttner, T., P. Jacobebbinghaus, and J. Ludsteck (2010). Occupational upgrading and the business cycle in west germany. *Economics - The Open-Access, Open-Assessment E-Journal* 4(10), 1–37.
- Burnside, C. and M. Eichenbaum (1996, December). Factor-hoarding and the propagation of businesscycle shocks. *American Economic Review* 86(5), 1154–74.
- Campbell III, C. M. (1997). The variation in wage rigidity by occupation and union status in the us. Oxford Bulletin of Economics and Statistics 59(1), 133–147.

- Chassamboulli, A. (2011). Cyclical upgrading of labor and employment differences across skill groups. The B.E. Journal of Macroeconomics 11(1).
- Clark, K. B. and L. H. Summers (1981). Demographic differences in cyclical employment variation. Journal of Human Resources 16(1), 61–79.
- Davis, S. J., R. J. Faberman, and J. Haltiwanger (2006, Summer). The flow approach to labor markets: New data sources and micro-macro links. *Journal of Economic Perspectives* 20(3), 3–26.
- Devereux, P. J. (2000, January). Task assignment over the business cycle. Journal of Labor Economics 18(1), 98–124.
- Devereux, P. J. (2004, January). Cyclical quality adjustment in the labor market. Southern Economic Journal 70(3), 600–615.
- Dolado, J., M. Jansen, and J. Jimeno (2009, 01). On-the-job search in a matching model with heterogeneous jobs and workers. *Economic Journal 119*(534), 200–228.
- Elsby, M., B. Hobjin, and A. Sahin (2010). The labor market in the great recession. Working Paper Series 2010-07, Federal Reserve Bank of San Francisco.
- Elsby, M. W. L., R. Michaels, and G. Solon (2009, January). The ins and outs of cyclical unemployment. American Economic Journal: Macroeconomics 1(1), 84–110.
- Evans, P. (1999). Occupational downgrading and upgrading in britain. *Economica* 66(261), pp. 79–96.
- Fair, R. C. (1985, March). Excess labor and the business cycle. American Economic Review 75(1), 239–45.
- Fay, J. A. and J. L. Medoff (1985, September). Labor and output over the business cycle: Some direct evidence. American Economic Review 75(4), 638–55.
- Firpo, S., N. M. Fortin, and T. Lemieux (2011, February). Occupational tasks and changes in the wage structure. IZA Discussion Papers 5542, Institute for the Study of Labor (IZA).
- Fujita, S. and G. Ramey (2009, 05). The cyclicality of separation and job finding rates. International Economic Review 50(2), 415–430.
- Galeotti, M., L. J. Maccini, and F. Schiantarelli (2005, April). Inventories, employment and hours. Journal of Monetary Economics 52(3), 575–600.

- Gathmann, C. and U. Schönberg (2010, 01). How general is human capital? a task-based approach. Journal of Labor Economics 28(1), 1–49.
- Geel, R. and U. Backes-Gellner (2009, September). Occupational mobility within and between skill clusters: An empirical analysis based on the skill-weights approach. Economics of Education Working Paper Series 0047, University of Zurich, Institute for Strategy and Business Economics (ISU).
- Goos, M. and A. Manning (2007, February). Lousy and lovely jobs: The rising polarization of work in britain. *The Review of Economics and Statistics* 89(1), 118–133.
- Hamermesh, D. S. (1995). Labour demand and the source of adjustment costs. The Economic Journal 105(430), pp. 620–634.
- Hijzen, A., R. Upward, and P. W. Wright (2010). The income losses of displaced workers. Journal of Human Resources 45(1).
- Hoynes, H. (1999, June). The employment, earnings, and income of less skilled workers over the business cycle. NBER Working Papers 7188, National Bureau of Economic Research, Inc.
- Hoynes, H., D. L. Miller, and J. Schaller (2012, Summer). Who suffers during recessions? Journal of Economic Perspectives 26(3), 27–48.
- Ingram, B. F. and G. R. Neumann (2006, February). The returns to skill. *Labour Economics* 13(1), 35–59.
- Jacobson, L. S., R. J. LaLonde, and D. G. Sullivan (1993, September). Earnings losses of displaced workers. American Economic Review 83(4), 685–709.
- Jaimovich, N. and H. E. Siu (2009, June). The young, the old, and the restless: Demographics and business cycle volatility. American Economic Review 99(3), 804–26.
- Jaimovich, N. and H. E. Siu (2012, August). The trend is the cycle: Job polarization and jobless recoveries. NBER Working Papers 18334, National Bureau of Economic Research, Inc.
- James, J. (2011). Ability matching and occupational choice. Working Paper 1125, Federal Reserve Bank of Cleveland.
- Kambourov, G. and I. Manovskii (2009, 02). Occupational specificity of human capital. International Economic Review 50(1), 63–115.

- Keane, M. and E. Prasad (1993, December). Skill levels and the cyclical variability of employment, hours, and wages. *IMF Staff Papers* 40(4), 711–743.
- Khalifa, S. (2009, April). Heterogeneous workers and occupations: Inequality, unemployment, and crowding out. Southern Economic Journal 75(4), 1141–1164.
- Kydland, F. E. (1984, January). Labor-force heterogeneity and the business cycle. Carnegie-Rochester Conference Series on Public Policy 21(1), 173–208.
- Lerman, R. I., S.-M. McKernan, and S. Riegg. (2004). The scope of employer-provided training in the united states: Who, what, where, and how much? In R. A. S. Christopher J. O'Leary and S. A. Wandner (Eds.), *Job Training Policy in the United States*. W.E. Upjohn Institute for Employment Research.
- Leuven, E. (2005, 02). The economics of private sector training: A survey of the literature. *Journal* of Economic Surveys 19(1), 91–111.
- Lillard, L. A. and H. W. Tan (1986). Private sector training: Who gets it and what are its effects? Technical report, RAND, Santa Monica, CA.
- Liu, T. and L. C. Spector (2005, January). Dynamic employment adjustments over business cycles. *Empirical Economics* 30(1), 151–169.
- Manning, A. (2011). Imperfect Competition in the Labor Market, Volume 4 of Handbook of Labor Economics, Chapter 11, pp. 973–1041. Elsevier.
- Marchetti, D. J. and F. Nucci (2001, November). Labor effort over the business cycle. Temi di discussione (Economic working papers) 424, Bank of Italy, Economic Research and International Relations Area.
- Miller, R. L. (1971, March). The reserve labour hypothesis: Some tests of its implications. *Economic Journal* 81(321), 17–35.
- Moscarini, G. and F. Vella (2008, February). Occupational mobility and the business cycle. IZA Discussion Papers 3369, Institute for the Study of Labor (IZA).
- Oi, W. Y. (1962). Labor as a quasi-fixed factor. Journal of Political Economy 70, 538.

- Oyer, P. and S. Schaefer (2011). Chapter 20 personnel economics: Hiring and incentives. Volume 4, Part B of *Handbook of Labor Economics*, pp. 1769 – 1823. Elsevier.
- Platt, H. and M. B. P. Platt (2011). Revisiting the labor hoarding employment demand model: an economic order quantity approach. *Journal of Financial Transformation* 31, 158–163.
- Poletaev, M. and C. Robinson (2008, 07). Human capital specificity: Evidence from the dictionary of occupational titles and displaced worker surveys, 1984-2000. *Journal of Labor Economics* 26(3), 387–420.
- Reder, M. W. (1955). The theory of occupational wage differentials. The American Economic Review 45(5), pp. 833–852.
- Robinson, C. (2011). Occupational mobility, occupation distance and specific human capital. University of Western Ontario, CIBC Centre for Human Capital and Productivity Working Papers 20115, University of Western Ontario, CIBC Centre for Human Capital and Productivity.
- Shimer, R. (2012, April). Reassessing the ins and outs of unemployment. *Review of Economic Dynamics* 15(2), 127–148.
- Solon, G., W. Whatley, and A. H. Stevens (1997, April). Wage changes and intrafirm job mobility over the business cycle: Two case studies. *Industrial and Labor Relations Review* 50(3), 402–415.
- Teulings, C. N. (1993). The diverging effects of the business cycle on the expected duration of jobsearch. Oxford Economic Papers 45(3), pp. 482–500.
- Topel, R. H. (1982, September). Inventories, layoffs, and the short-run demand for labor. American Economic Review 72(4), 769–87.
- van Ours, J. C. and G. Ridder (1995). Job matching and job competition: Are lower educated workers at the back of the job queues? *European Economic Review 39*, 1717–1731.
- von Wachter, T., E. Handwerker, and A. Hildreth (2009, June). Estimating the "true" cost of job loss: Evidence using matched data from califormia 1991-2000. Working Papers 09-14, Center for Economic Studies, U.S. Census Bureau.
- Wen, Y. (2005). Labor hoarding and inventories. Working Papers 2005-040, Federal Reserve Bank of St. Louis.

Yamaguchi, S. (2012). Tasks and heterogeneous human capital. Journal of Labor Economics 30(1), pp. 1–53.

6 Appendix

6.1 Definition of occupational skills using the O*NET

I define four occupational skill measures using the Abilities section of the 17th edition of the O^*NET . In the abilities section²⁸ each occupation was evaluated by 8 analysts on 52 ability measures that may have influenced work performance in the occupation in 2006. The analysts evaluated on a scale of 1 to 5 how important the ability is in given occupations, and on a scale of 0 to 7 what level is needed to perform well. The level scales were anchored. For example at evaluating the level of oral comprehension, the anchor ran from "understand a television commercial" up to "understand a lecture on advanced physics", and at evaluating mathematical reasoning the anchor ran from "Determine how much 10 oranges will cost when they are priced at 2 for 20 cents" to "Determine the mathematics required to simulate a space craft landing on the moon".

The 52 ability measures were classified into 15 minor and 4 major categories:

- 1. Cognitive abilities
 - (a) Verbal abilities
 - (b) Idea generation and reasoning abilities
 - (c) Quantitative abilities
 - (d) Memory
 - (e) Perceptual abilities
 - (f) Spatial abilities
 - (g) Attentiveness
- 2. Psychomotor abilities

²⁸See, for example, at http://www.onetcenter.org/content.html.

- (a) Fine manipulative abilities
- (b) Control movement abilities
- (c) Reaction time and speed abilities
- 3. Physical abilities
 - (a) Physical strength abilities
 - (b) Endurance
 - (c) Flexibility, balance and coordination
- 4. Sensory abilities
 - (a) Visual abilities
 - (b) Auditory and speech abilities

In this project I did not use the sensory abilities, and I aggregated the rest of the information into 4 categories:

- 1. Non-routine cognitive abilities (From 1a to 1e)
- 2. Routine cognitive abilities (1f and 1g)
- 3. Psychomotor abilities
- 4. Physical abilities

Then the procedure I followed was the following:

- 1. I rescaled both the importance and the level scores to be between 0 and 1.
- 2. Then I aggregated the occupations to be the same as the most detailed occupations in the 2000 census, the 3-digit SOC codes.
- 3. Then I took the occupational means of the importance and level scores in the four aggregate ability measures.

- 4. Then I took the weighted average of these ability measures in my 191 occupation categories, where the weights were the number of workers in the occupations in the 2000 census.
- 5. Finally I standardized the occupational abilities in the 2000 census, and I created 10 deciles as well.

6.2 Deriving the marginal effect of GDP growth at various percentiles of the skill and education distribution

The estimated non-linear least squares model is

$$U_{it} = \exp\left(\beta_0 + \beta_1 d\ln GDP_t + \beta_2 S_i + \beta_3 d\ln GDP_t \times S_i\right) + u_{it}$$

The marginal effect of a 10% higher GDP growth is

$$\frac{1}{10}\frac{\partial E\left(U_{it}|S_{i},d\ln GDP_{t}\right)}{\partial d\ln GDP_{t}} = \frac{1}{10}\left(\beta_{1}+\beta_{3}S_{i}\right)\exp\left(\beta_{0}+\beta_{1}d\ln GDP_{t}+\beta_{2}S_{i}+\beta_{3}d\ln GDP_{t}\times S_{i}\right)$$

The marginal effect at various percentiles of the skill (or education) distribution and at mean GDP growth m_{gdp} is

$$\frac{1}{10}\frac{\partial E\left(U_{it}|S_{i}=S_{\tau},d\ln GDP_{t}=m_{gdp}\right)}{\partial d\ln GDP_{t}}=\frac{1}{10}\left(\beta_{1}+\beta_{3}S_{\tau}\right)\exp\left(\beta_{0}+\beta_{1}m_{gdp}+\beta_{2}S_{i}+\beta_{3}m_{gdp}\times S_{\tau}\right)$$

6.3 Derivations in the simple labor hoarding model

6.3.1 Equilibrium wages in a stable contract

There are two state variables: aggregate productivity $a_t \in \{a^B, a^R\}$ and whether the firm has a worker or not, $l \in \{0, 1\}$. The values in these states are:

$$V_{o}^{S}(a^{B},1) = (a^{B}s_{o} - w_{o}) + \beta p^{B}(1 - \gamma_{o})V_{o}^{S}(a^{B},1) + \beta p^{B}\gamma_{o}V_{o}^{S}(a^{B},0) + \beta p^{R}(1 - \gamma_{o})V_{o}^{S}(a^{R},1) + \beta p^{R}\gamma_{o}V_{o}^{S}(a^{R},0) V_{o}^{S}(a^{R},1) = (a^{R}s_{o} - w_{o}) + \beta p^{B}(1 - \gamma_{o})V_{o}^{S}(a^{B},1) + \beta p^{B}\gamma_{o}V_{o}^{S}(a^{B},0)$$

$$+\beta p^{R} (1 - \gamma_{o}) V_{o}^{S} (a^{R}, 1) + \beta p^{R} \gamma_{o} V_{o}^{S} (a^{R}, 0)$$

$$V_{o}^{S} (a^{B}, 0) = (a^{B} s_{o} - w_{o} - h_{o}) + \beta p^{B} (1 - \gamma_{o}) V_{o}^{S} (a^{B}, 1) + \beta p^{B} \gamma_{o} V_{o}^{S} (a^{B}, 0)$$

$$+\beta p^{R} (1 - \gamma_{o}) V_{o}^{S} (a^{R}, 1) + \beta p^{R} \gamma_{o} V_{o}^{S} (a^{R}, 0)$$

$$V_{o}^{S} (a^{R}, 0) = 0 + \beta p^{B} V_{o}^{S} (a^{B}, 0) + \beta p^{R} V_{o}^{S} (a^{R}, 0)$$

Thus

$$V_o^S(a^B, 1) - V_o^S(a^B, 0) = h_o$$
(5)

Free entry among firms assures that the value of hiring a worker is zero, which implies that

$$\begin{aligned} V_o^S \left(a^B, 0 \right) &= 0 \\ V_o^S \left(a^R, 0 \right) &= 0 \\ V_o^S \left(a^B, 1 \right) &= \left(a^B s_o - w_o^S \right) + \beta p^B \left(1 - \gamma_o \right) V_o^S \left(a^B, 1 \right) + \beta p^R \left(1 - \gamma_o \right) \left(V_o^S \left(a^B, 1 \right) - \left(a^B - a^R \right) s_o \right) \\ &= \frac{\left(a^B s_o - w_o^S \right) - \beta p^R \left(1 - \gamma_o \right) \left(a^B - a^R \right)}{1 - \beta \left(1 - \gamma_o \right)} \\ V_o^S \left(a^R, 1 \right) &= \frac{\left(a^R s_o - w_o^S \right) - \beta p^B \left(1 - \gamma_o \right) \left(a^B - a^R \right) s_o}{1 - \beta \left(1 - \gamma_o \right)} \end{aligned}$$

Using (5) the equilibrium wage is

$$h_{o} = \frac{\left(a^{B}s_{o} - w_{o}^{S}\right) - \beta p^{R} \left(1 - \gamma_{o}\right) \left(a^{B} - a^{R}\right)}{1 - \beta \left(1 - \gamma_{o}\right)}$$
$$w_{o}^{S} = a^{B}s_{o} - \beta p^{R} \left(1 - \gamma_{o}\right) \left(a^{B} - a^{R}\right) s_{o} - h_{o} \left(1 - \beta \left(1 - \gamma_{o}\right)\right)$$

In order to compute the expected wage of a worker we need the probability that he is employed. In booms everyone is employed. In recessions people are only employed if they were employed at the last time and they did not quit exogenously:

$$P_o^S(E) = p^B + p^R (1 - \gamma_o) P_o^S(E)$$
$$= \frac{p^B}{1 - p^R (1 - \gamma_o)}$$

The expected wage of a worker

$$W_{o}^{S} = \frac{p^{B}}{1 - p^{R} (1 - \gamma_{o})} \left[a^{B} s_{o} - \beta p^{R} (1 - \gamma_{o}) \left(a^{B} - a^{R} \right) s_{o} - h_{o} \left(1 - \beta \left(1 - \gamma_{o} \right) \right) \right]$$

6.3.2 Equilibrium wages in an unstable contract

The values in different states are

$$\begin{split} V_{o}^{U}\left(a^{B},1\right) &= \left(a^{B}s_{o}-w_{o}\right)+\beta p^{B}\left(1-\gamma_{o}\right)V_{o}^{U}\left(a^{B},1\right)+\beta p^{B}\gamma_{o}V_{o}^{U}\left(a^{B},0\right)\\ &+\beta p^{R}\left(1-\gamma_{o}\right)V_{o}^{U}\left(a^{R},1\right)+\beta p^{R}\gamma_{o}V_{o}^{U}\left(a^{R},0\right)\\ V_{o}^{U}\left(a^{R},1\right) &= -f_{o}+\beta p^{R}V_{o}^{U}\left(a^{R},0\right)+\beta p^{B}V_{o}^{U}\left(a^{B},0\right)\\ V_{o}^{U}\left(a^{B},0\right) &= \left(a^{B}s_{o}-w_{o}-h_{o}\right)+\beta p^{B}\left(1-\gamma_{o}\right)V_{o}^{U}\left(a^{B},1\right)+\beta p^{B}\gamma_{o}V_{o}^{U}\left(a^{B},0\right)\\ &+\beta p^{R}\left(1-\gamma_{o}\right)V_{o}^{U}\left(a^{R},1\right)+\beta p^{R}\gamma_{o}V_{o}^{U}\left(a^{R},0\right)\\ V_{o}^{U}\left(a^{R},0\right) &= \beta p^{R}V_{o}^{F}\left(a^{R},0\right)+\beta p^{B}V_{o}^{F}\left(a^{B},0\right) \end{split}$$

Thus

$$V_{o}^{S}\left(a^{B},1\right) - V_{o}^{S}\left(a^{B},0\right) = h_{o}$$
(6)

Free entry among firms assures that the value of hiring a worker is zero, which implies that

$$\begin{split} V_o^U \left(a^B, 0 \right) &= 0 \\ V_o^U \left(a^R, 0 \right) &= 0 \\ V_o^U \left(a^R, 1 \right) &= -f_o \\ V_o^U \left(a^B, 1 \right) &= \left(a^B s_o - w_o^U \right) + \beta p^B \left(1 - \gamma_o \right) V_o^F \left(a^B, 1 \right) - \beta p^R \left(1 - \gamma_o \right) f_o \\ &= \frac{\left(a^B s_o - w_o^U \right) - \beta p^R \left(1 - \gamma_o \right) f_o}{1 - \beta p^B \left(1 - \gamma_o \right)} \end{split}$$

Using (6) the equilibrium wage is

$$h_{o} = \frac{\left(a^{B}s_{o} - w_{o}^{U}\right) - \beta p^{R}\left(1 - \gamma_{o}\right)f_{o}}{1 - \beta p^{B}\left(1 - \gamma_{o}\right)}$$
$$w_{o}^{U} = a^{B}s_{o} - \beta p^{R}\left(1 - \gamma_{o}\right)f_{o} - h_{o}\left(1 - \beta p^{B}\left(1 - \gamma_{o}\right)\right)$$

Workers are only employed in booms:

$$P_o^U(E) = p^B$$

And thus, the expected wage is

$$W_{o}^{U} = p^{B} \left(a^{B} s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) f_{o} - h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o} \right) \right) \right)$$

6.4 Proof that the unstable contract is never socially optimal

I prove by contradiction that if $W_o^U > 0$ then $W_o^S > W_o^U$. Let's assume that $W_o^U > 0$ and $W_o^S \le W_o^U$. The first condition sequentially imply that

$$\begin{split} W_{o}^{U} &> 0\\ a^{B}s_{o} &> \beta p^{R} \left(1 - \gamma_{o}\right) f_{o} + h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o}\right)\right) > h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o}\right)\right)\\ \left(1 - \beta\right) a^{B}s_{o} &> (1 - \beta) h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o}\right)\right) = h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o}\right) - \beta + \beta p^{B} \left(1 - \gamma_{o}\right)\right)\\ &> h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o}\right) - \beta\right) = h_{o} \left(1 - \beta \left(1 + \beta p^{B} \left(1 - \gamma_{o}\right)\right)\right) \end{split}$$

and thus

$$(1-\beta) a^B s_o > h_o \left(1-\beta \left(1+\beta p^B \left(1-\gamma_o\right)\right)\right)$$
(7)

The second condition sequentially imply that

$$0 \leq W_{o}^{U} - W_{o}^{S}$$

$$\leq p^{B} \left(a^{B} s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) f_{o} - h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o} \right) \right) \right)$$

$$- \frac{p^{B}}{1 - p^{R} \left(1 - \gamma_{o} \right)} \left[a^{B} s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) \left(a^{B} - a^{R} \right) s_{o} - h_{o} \left(1 - \beta \left(1 - \gamma_{o} \right) \right) \right] \quad (8)$$

$$0 \leq \left(1 - p^{R} \left(1 - \gamma_{o} \right) \right) \left(a^{B} s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) f_{o} - h_{o} \left(1 - \beta p^{B} \left(1 - \gamma_{o} \right) \right) \right)$$

$$- a^{B} s_{o} - \beta p^{R} \left(1 - \gamma_{o} \right) \left(a^{B} - a^{R} \right) s_{o} - h^{o} \left(1 - \beta \left(1 - \gamma_{o} \right) \right)$$

$$(9)$$

$$0 \leq h_{o} \left[\left(1 - \beta \left(1 - \gamma_{o} \right) \right) - \left(1 - \beta p^{B} \left(1 - \gamma_{o} \right) \right) + p^{R} \left(1 - \gamma_{o} \right) \left(1 - \beta p^{B} \left(1 - \gamma_{o} \right) \right) \right] - \left(1 - p^{R} \left(1 - \gamma_{o} \right) \right) \beta p^{R} \left(1 - \gamma_{o} \right) f_{o} - p^{R} \left(1 - \gamma_{o} \right) \left(a^{B} \left(1 - \beta \right) + a^{R} \beta \right) s_{o}$$
(10)
$$0 \leq h_{o} \left(1 - \beta \left(1 + p^{B} \left(1 - \gamma_{o} \right) \right) \right) - \left(1 - p^{R} \left(1 - \gamma_{o} \right) \right) \beta f_{o} - \left(a^{B} \left(1 - \beta \right) + a^{R} \beta \right) s_{o}$$

$$(1-\beta) a^B s_o < h_o \left(1-\beta \left(1+p^B \left(1-\gamma_o\right)\right)\right)$$

$$(11)$$

(7) and (11) contradict each other and thus the original claim is proved.

6.5 Derivation of the market outcome

A stable contract is the market outcome if the per-period wage is higher in the stable contract

$$w_o^S \geq w_o^U$$

$$a^B s_o - \beta p^R (1 - \gamma_o) \left(a^B - a^R\right) s_o - h_o \left(1 - \beta \left(1 - \gamma_o\right)\right) \geq a^B s_o - \beta p^R \left(1 - \gamma_o\right) f_o - h_o \left(1 - \beta p^B \left(1 - \gamma_o\right)\right)$$

$$\beta p^R (1 - \gamma_o) h_o + \beta p^R \left(1 - \gamma_o\right) f_o \geq \beta p^R \left(1 - \gamma_o\right) \left(a^B - a^R\right) s_o$$

$$h_o + f_o \geq \left(a^B - a^R\right) s_o$$

6.6 Tables and figures

	Е	W	NC	RC	PM	\mathbf{PS}
Avg. education, E	1.00					
Avg. wage, W	0.78	1.00				
Non-routine cognitive abilities, NC	0.80	0.71	1.00			
Routine cognitive abilities, RC	-0.03	0.29	0.21	1.00		
Psychomotor abilities, PM	-0.58	-0.25	-0.47	0.62	1.00	
Physical abilities, PS	-0.61	-0.39	-0.57	0.43	0.82	1.00

Table 5: Correlations between 7 alternative occupational skill variables

*Occupational skills are all based on averages in the 2000 census in each of my 191 occupations.; N=191

		Panel A									
	0	Occupational skill deciles (avg. yrs. of educ. in 191 occupations)									
Major occupation groups	1	2	3	4	5	6	7	8	9	10	Total
Management and related	0.0	0.0	0.0	0.0	0.0	0.0	0.0	50.7	49.2	8.5	11.4
Professional	0.0	0.0	0.0	0.0	0.0	0.0	3.0	17.2	38.8	91.5	15.1
Technicians	0.0	0.0	0.0	0.0	0.0	0.0	16.8	8.7	9.3	0.0	3.2
Sales	0.0	0.0	24.1	1.7	0.0	39.4	33.5	8.6	2.7	0.0	11.5
Office and admin. support	0.0	0.0	10.1	6.9	40.1	56.2	41.8	7.5	0.0	0.0	16.7
Production and repair	3.7	21.3	23.1	34.6	27.7	0.8	2.6	0.0	0.0	0.0	11.0
Operators and laborers	47.2	71.3	25.6	13.8	8.0	0.0	0.0	0.0	0.0	0.0	16.2
Protective service	0.0	0.0	1.5	0.0	5.4	2.2	2.4	7.3	0.0	0.0	2.0
Food prep. and cleaning	49.1	7.4	15.6	8.8	2.5	0.0	0.0	0.0	0.0	0.0	8.2
Personal services	0.0	0.0	0.0	34.2	16.2	1.4	0.0	0.0	0.0	0.0	4.8
Total	100	100	100	100	100	100	100	100	100	100	100

 Table 6: Distribution of major occupational groups within occupational skill deciles, weighted 2000

 census values

	Panel B										
	Occu	Occupational skill deciles (avg. ln weakly wage in 191 occupations)									
Major occupation groups	1	2	3	4	5	6	7	8	9	10	Total
Management and related	0.0	0.0	0.0	0.0	0.0	0.0	1.7	15.4	20.6	64.9	11.4
Professional	0.0	3.7	0.0	2.2	4.7	19.1	6.4	31.7	53.2	26.3	15.1
Technicians	0.0	0.0	0.0	0.0	11.0	4.5	5.3	6.3	0.0	6.1	3.2
Sales	24.9	0.0	1.5	0.0	0.0	0.0	74.6	3.3	6.0	2.7	11.5
Office and admin. support	6.9	15.6	42.4	62.3	5.8	23.4	0.0	8.6	0.0	0.0	16.7
Production and repair	0.0	2.1	3.6	6.2	40.9	16.3	10.3	31.4	2.6	0.0	11.0
Operators and laborers	3.8	29.7	35.7	22.9	37.5	34.1	1.8	0.6	0.8	0.0	16.2
Protective service	0.0	0.0	7.1	0.0	0.0	2.7	0.0	2.8	6.8	0.0	2.0
Food prep. and cleaning	53.5	19.5	6.0	1.7	0.0	0.0	0.0	0.0	0.0	0.0	8.2
Personal services	11.0	29.6	3.7	4.7	0.0	0.0	0.0	0.0	0.0	0.0	4.8
Total	100	100	100	100	100	100	100	100	100	100	100

	[1]	[2]	[3]	[4]
dln(GDP), DG	-8.757	-6.644	-1.419	-3.418
	$[0.139]^{**}$	$[0.320]^{**}$	$[0.434]^{**}$	$[0.560]^{**}$
Occupational skill, SE	-0.631			-0.499
	$[0.004]^{**}$			$[0.004]^{**}$
SE X DG	-1.666			-1.44
	$[0.135]^{**}$			$[0.146]^{**}$
Years of education, E		-0.102	-0.1	-0.04
		$[0.001]^{**}$	$[0.001]^{**}$	[0.001]**
E X DG		-0.109	-0.118	-0.084
		$[0.027]^{**}$	$[0.027]^{**}$	[0.034]*
Age, A			-0.024	-0.019
			$[0.000]^{**}$	$[0.000]^{**}$
A X DG			-0.137	-0.123
			$[0.008]^{**}$	[0.007]**
Female, F			-0.231	-0.152
			$[0.006]^{**}$	[0.005]**
F X DG			2.763	3.227
			$[0.188]^{**}$	$[0.179]^{**}$
White, W			-0.472	-0.425
			$[0.006]^{**}$	$[0.006]^{**}$
W X DG			-2.305	-1.967
			$[0.191]^{**}$	$[0.181]^{**}$
Constant	-2.762	-1.291	0.033	-1.134
	$[0.004]^{**}$	$[0.010]^{**}$	$[0.013]^*$	[0.017]**
N	2462447	2462447	2462447	2462447
R squared	0.082	0.071	0.082	0.091
Marginal effect of 10% higher	GDP grow	th on unem	ployment at	various percentiles
of the skill distribution (mode	el 1) and th	e education	distribution	(model 2)
5th percentile	-0.08	-0.069		
	$[0.001]^{**}$	$[0.001]^{**}$		
25th percentile	-0.063	-0.053		
	$[0.001]^{**}$	$[0.001]^{**}$		
Median	-0.044	-0.048		
	$[0.001]^{**}$	$[0.001]^{**}$		
75th percentile	-0.03	-0.036		
	$[0.001]^{**}$	$[0.001]^{**}$		
95th percentile	-0.016	-0.033		
	$[0.000]^{**}$	$[0.001]^{**}$		
Ratio: 5th / 95th percentile	5.03	2.077		
	$[0.202]^{**}$	[0.052]**		

Table 7: Non-linear least squares estimates of unemployment, March CPS, 1983-2012

*The right hand side of the models is the exponent of an index created by the linear combination of the variables in the model; dln(GDP) is defined as the log difference between GDP in the previous calendar year and the year before that; Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations standardized to have zero mean and standard deviation 1 in the 2000 census; The marginal effects are computed at the mean GDP growth in the sample which is 2.5 percent.

	······································	., ., .,		
	Males	Females	Whites	Non-whites
dln(GDP), DG	-0.6	-0.365	-0.472	-0.546
	$[0.011]^{**}$	$[0.010]^{**}$	$[0.008]^{**}$	$[0.022]^{**}$
Occupational skill, SE	-0.04	-0.029	-0.033	-0.045
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.001]^{**}$
SE X DG	0.243	0.065	0.179	0.133
	$[0.010]^{**}$	$[0.011]^{**}$	$[0.008]^{**}$	$[0.022]^{**}$
Constant	0.08	0.067	0.069	0.101
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.001]^{**}$
Ν	1302531	1159916	2077104	385343
R squared	0.023	0.015	0.018	0.023

Table 8: OLS regressions of unemployment, March CPS, 1983-2012

Marginal effect of 10% higher GDP growth on unemployment at various percentiles of the skill distribution

1				
5th percentile	-0.092	-0.045	-0.071	-0.072
	$[0.002]^{**}$	$[0.002]^{**}$	$[0.001]^{**}$	$[0.004]^{**}$
25th percentile	-0.078	-0.041	-0.06	-0.064
	[0.001]**	$[0.001]^{**}$	[0.001]**	$[0.003]^{**}$
Median	-0.06	-0.036	-0.047	-0.055
	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.002]^{**}$
75th percentile	-0.042	-0.032	-0.034	-0.045
	[0.001]**	$[0.001]^{**}$	[0.001]**	$[0.003]^{**}$
95th percentile	-0.011	-0.024	-0.011	-0.028
	$[0.002]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$	$[0.005]^{**}$
Ratio: 5th / 95th percentile	8.054	1.907	6.15	2.571
	$[1.706]^{**}$	$[0.228]^{**}$	$[0.944]^{**}$	$[0.537]^{**}$
	Age 16-29	Age 30-39	Age 40-49	Age 50-64
dln(GDP), DG	-0.579	-0.506	-0.483	-0.486
	[0.017]**	$[0.014]^{**}$	$[0.013]^{**}$	$[0.014]^{**}$
Occupational skill, SE	-0.051	-0.034	-0.028	-0.023
	[0.001]**	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$
SE X DG	0.213	0.187	0.184	0.152
	$[0.019]^{**}$	$[0.013]^{**}$	$[0.013]^{**}$	$[0.013]^{**}$
Constant	0.096	0.072	0.062	0.059
	[0.001]**	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$
N	674346	654536	618760	514805
R squared	0.022	0.02	0.016	0.012
Marginal effect of 10% higher	· GDP growt	h on unemple	oyment	
at various percentiles of the s	kill distribut	ion		
5th percentile	-0.086	-0.075	-0.072	-0.068
	[0.003]**	$[0.002]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$
25th percentile	-0.074	-0.064	-0.062	-0.06
	$[0.002]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$
Median	-0.058	-0.051	-0.048	-0.049
	$[0.002]^{**}$	$[0.001]^{**}$	[0.001]**	$[0.001]^{**}$
75th percentile	-0.042	-0.036	-0.034	-0.037
-	$[0.003]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$	$[0.002]^{**}$
95th percentile	-0.015	-0.013	-0.011	-0.018

 $[1.750]^{**}$ *dln(GDP) is defined as the log difference between GDP in the previous calendar year and the year before that; Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations standardized to have zero mean and standard deviation 1 $\stackrel{47}{\text{in}}$ the 2000 census

 $[0.005]^{**}$

5.63

Ratio: 5th / 95th percentile

 $[0.003]^{**}$

5.657

 $[1.300]^{**}$

 $[0.003]^{**}$

6.318

[1.601]**

 $[0.003]^{**}$

3.74

 $[0.640]^{**}$

	\mathbf{E}	W	NC	RC	\mathbf{PM}	\mathbf{PS}			
dln(GDP), DG	-0.5	-0.48	-0.49	-0.471	-0.489	-0.473			
	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$			
Occupational skill, SE	-0.036	-0.025	-0.034	0.006	0.029	0.03			
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$			
SE X DG	0.176	0.067	0.147	-0.108	-0.191	-0.153			
	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$	$[0.007]^{**}$			
Constant	0.074	0.074	0.074	0.072	0.073	0.073			
	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$	$[0.000]^{**}$			
N	2462447	2462447	2462447	2462447	2462447	2462447			
R squared	0.02	0.012	0.018	0.002	0.012	0.014			
Marginal effect of 10% higher GDP growth on unemployment									
at various percentiles of the s	kill distribu	tion							
5th percentile	-0.073	-0.059	-0.077	-0.029	-0.021	-0.025			
	$[0.001]^{**}$	$[0.001]^{**}$	$[0.002]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	[0.001]**			
25th percentile	-0.063	-0.052	-0.058	-0.039	-0.033	-0.035			
	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	[0.001]**			
Median	-0.05	-0.048	-0.05	-0.047	-0.047	-0.047			
	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	[0.001]**			
75th percentile	-0.037	-0.043	-0.036	-0.053	-0.064	-0.06			
	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	[0.001]**			
95th percentile	-0.015	-0.037	-0.028	-0.067	-0.079	-0.069			
	$[0.002]^{**}$	$[0.001]^{**}$	$[0.001]^{**}$	$[0.002]^{**}$	$[0.001]^{**}$	[0.001]**			
Ratio: 5th / 95th percentile	4.941	1.635	3.499	0.482	0.275	0.351			
	$[0.579]^{**}$	$[0.096]^{**}$	$[0.320]^{**}$	$[0.025]^{**}$	$[0.017]^{**}$	$[0.020]^{**}$			

Table 9: OLS regressions of unemployment by alternative skill measures, March CPS, 1983-2012

*dln(GDP) is defined as the log difference between GDP in the previous calendar year and the year before that; Different columns are based on difference occupational skill measures. E: average years of education in the 2000 census; W: average wage of full time full year workers in the 2000 census; NC: Non-routine cognitive abilities in the O*NET; RC: routine cognitive abilities in the O*NET; PM: psychomotor abilities in the O*NET; PS: Physical abilities in the O*NET Figure 6: Unemployment rate by the skill level of last year's occupation in different periods, March CPS 1983-2012



Panel A: In major occupation groups

Panel B: By occupational skill deciles*



*Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census by dividing the population into 10 roughly equal groups based on their occupations. Higher number means higher skilled occupation. In each occupational group there are 6 bars representing 6 time periods with the ones on the right being the more recent.





Panel B: Skill is average years of education in 1990 census in 383 occupations





Figure 8: Unemployment rate by occupational skills in different periods, March CPS 1983-2012 Panel A: Non-routine cognitive skills in 191 occupations

Panel B: Routine cognitive skills in 191 occupations



Figure 9: Unemployment rate by occupational skills in different periods, March CPS 1983-2012 Panel A: Psychomotor skills in 191 occupations



Panel B: Physical skills in 191 occupations



Figure 10: Unemployment rates by gender and the skill decile of the last occupation*, March CPS 1983-2012



*Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census, and higher number means higher skilled occupation. The gray areas indicate NBER recession dates.



Figure 11: Average unemployment rate by occupation, March CPS, 2006-2012, colored by 10 major groups.

Panel B: 2006-2007





Figure 12: Evolution of job-loss and job-finding probabilities by major occupational groups, 1996-2013 Panel A: job-loss probabilities

Panel B: job-finding probabilities



*The figure shows yearly moving averages of monthly job-loss and job-finding probabilities using the monthly CPS. For the unemployed, the occupation at the last job is used. The gray areas indicate NBER recession dates.



Figure 13: Industry GDP growth and skills Panel A: Standard deviation of industry GDP growth, 1983-2012





*Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census, and higher number means higher skilled occupation.



Figure 14: Average hours of work last week by occupational groups*, March CPS 1983-2012 Panel A: In major occupational groups

Panel B: In occupational skill deciles



*Conditional on working positive hours last week. Occupational skill is defined as the average years of education in the 2000 census in each of my 191 occupations. Deciles are taken in the 2000 census by dividing the population into 10 roughly equal groups based on their occupations. Higher number means higher skilled occupation.